Author's response to reviews

Title: Predictors of mortality among elderly dependent home care patients

Authors:

Joan Gene Badia (jgene@clinic.ub.es)
Alicia Borràs Santos (aborras@creal.cat)
Joan Carles Contel Segura (jccontel@ambitcp.catsalut.net)
Carlos Ascaso Terén (carlosascaso@ub.edu)
Laura Corredoira González (laura.corredoira@gmail.com)
Ester Limón (30705elr@comb.es)
Pedro Gallo de Puelles (pgallodep@ub.edu)
HC>65 Research Team (joangenebadia@gmail.com)

Version: 2 Date: 9 July 2012

Author's response to reviews:

Editor

BMC Health Services Research

6th, July 2012

Dear Editor,

Please find, attached, our manuscript entitled “Predictors of mortality among elderly dependent home care patients” which we would like considered for publication in BMC Health Services Research once we introduced the changes mentioned by you in the email:

1.- I apologize to mention the Scandinavian Journal in the letter. The mistake is due to I copied and pasted the letter to the editor already used for another paper, and I did not corrected everything.

2.- The paper is not already submitted to another journal and has been previously submitted to the British Journal of General Practice and rejected because the topic was not exclusively related to the UK. Some parts of the project such is the method section were previously published. The references of these previous publications related to this paper are mentioned in the bibliography section.

3.- We added the competing interests and the author’s contribution after the conclusions section.
Yours sincerely,

Joan Gené-Badia, MD, PhD
Universitat de Barcelona
Consorci d’Atenció Primaria de Salut de l’Eixample (CAPSE)
C/ Roselló 161
08036 Barcelona,
Spain
Tel: (+34) 93 227 98 55; Fax: (+34) 93 227 98 05
e-mail: jgene@clinic.ub.es